Author's response to reviews

Title: Treatment of two postoperative endophthalmitis cases due to Aspergillus flavus and Scopulariopsis spp. with local and systemic antifungal therapy

Authors:

Sayime Aydin (sayimeaydin@mynet.com)
Bulent Ertugrul (mbulentertugrul@yahoo.com)
Berna Gultekin (berngul@yahoo.com)
Guliz Uyar (gulizuyar@yahoo.com)
Erkin Kir (erkinkir@yahoo.com)

Version: 7 Date: 3 July 2007

Author's response to reviews: see over
Response to reviewer;

General compulsory revisions:

1. The sentences were changed as follows:
   “…… and IV voriconazole and caspofungin.”, In the abstract.
   “……amikacin (0.4mg/0.1cc) and vancomycin (1mg/0.1cc) were given intravitreally.”, on page 3.
   “…… were cultured perioperatively, IOL was also cultured.”, on page 3.
   “Vitreous was clean.”, on page 4.
   “….. fortified gentamicin and vancomycin ophthalmic solutions,……”, on page 4.
   “….. catheterization, genito-urinary and dental interventions increase the risk of endophthalmitis [12-15].”, on page 5.
   “….. counting…..”, on page 2 and on page 5.
   “Henceforth, IV amphotericin-B was stopped and IV voriconazole (4 mg/kg, b.i.d) and caspofungin (50mg/day) were started.”, on page 4.

Bulent Ertugrul (MD)
Adnan Menderes University Medical Faculty
Department of Infectious Diseases and Clinical Microbiology
Aydin / Turkey
tel #: 090 532 645 66 21
fax #: 090 256 214 64 95